Author's response to reviews

Title: Spontaneous regression of congenital epulis: a case report and review of literature

Authors:

Priyanshi Ritwik (pritwi@lsuhsc.edu)
Robert B Brannon (rbrann@lsuhsc.edu)
Robert J Musselman (rmusse1@mac.com)

Version: 4 Date: 23 July 2010

Author's response to reviews:

Major revisions

Author list - names of all authors have been changed to the following prescribed format: forename, initial, surname

Affiliations - Each affiliation has been written in the following format: department, institute, city, and country.

Patient information: ethnicity of the patient (African American) has been added in the abstract.

Author contributions: all authors have read and approved the final manuscript.

Minor revisions

Title - has been changed to sentence case

Manuscript format: coloured highlighting and comments relating to reviewer corrections have been removed